RESEARCH

Journal of Eating Disorders

Open Access

Exploration of interoceptive capabilities in avoidant/restrictive food intake disorder and anorexia nervosa



Nandini Datta^{1*} and James D. Lock¹

Abstract

Objective This proof-of-concept study explores the role of aberrant interoception as a possible mechanism underlying restrictive eating symptoms in avoidant/restrictive food intake disorder (ARFID) compared to anorexia nervosa (AN) and healthy comparisons (HC).

Method We report preliminary normative adolescent interoceptive data in HCs (n = 100) compared to adolescents with ARFID (n = 30) and AN (N = 23). Adolescents (12–18) participated in a one-time virtual visit to assess heartrate guessing accuracy (interoceptive accuracy), correlation between confidence in heartrate guess and accuracy (interoceptive awareness), and self-reported interoception (interoceptive sensibility).

Results HC adolescents had comparable interoceptive outcomes relative to published adult norms, consistent with existing literature. Data suggest that adolescents with ARFID have poor heartbeat guessing accuracy and experience challenges deciphering interoceptive signals, possibly contributing to symptoms. While adolescents with AN have *greater* heartbeat guessing accuracy, they cite difficulty trusting body cues, perhaps contributing to their lack of confidence in interoceptive cue detection.

Conclusions Preliminary results reflect differences in interoception between the three groups.

Keywords Interoception, Adolescent, ARFID, Anorexia nervosa

Plain English summary

In this study, we wanted to understand how individuals with restrictive eating disorders, like avoidant/restrictive food intake disorder (ARFID) and anorexia nervosa (AN), sense and respond to different signals in their body, such as heart-beat. This process is called interoception. Difficulty sensing body signals can lead to various mental health disorders, such as eating disorders. We asked adolescent participants without an eating disorder and adolescents with ARFID and AN to guess their heartbeat while wearing a finger pulse oximeter, to track interoceptive accuracy (or how accurate their heartbeat guesses were). We also measured their self-reported sense of interoception and overall metacognitive awareness of their ability to accurately interpret body signals. Our data replicate findings in existing healthy teenagers. For teens with eating disorders, those with ARFID had difficulty with interoceptive accuracy and listening to their bodies, but they had good awareness overall of their accuracy. Teens with AN had better accuracy but were more unsure about trusting their body and body signaling. Overall, our initial findings suggest that how individuals

*Correspondence: Nandini Datta nandinid@stanford.edu Full list of author information is available at the end of the article



© The Author(s) 2023. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.gr/licenses/by/4.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.gr/licenses/by/4.0/. The Creative Commons Public Domain Dedicated in a credit line to the data.

with eating disorders sense, interpret, and respond to body signals might help explain their eating habits. More research is needed in this area.

Background

Avoidant/restrictive food intake disorder (ARFID) is an eating disorder (ED) characterized by restrictive eating habits resulting in nutritional deficiencies and/or psychosocial impairment [1, 2]. Unlike anorexia nervosa (AN), there are no accompanying concerns about body shape or weight; rather, restrictive eating is driven by low appetite, severe "picky" eating due to sensory sensitivity, or fear of aversive consequences [3]. ARFID has prevalence rates from a specialized eating disorder service at 5-22%, and from non-clinical samples at 0.3-15.5% [4, 5], making it a serious mental health concern. Limited published studies have investigated the underlying pathophysiology of ARFID, which is necessary to better understand this debilitating disorder and inform treatments. Both ARFID and AN share sensory disturbances in gastric signaling, such as post-prandial discomfort and early satiety [6]. To that end, one mechanism implicated in AN that may impact symptom development in ARFID is aberrant interoception, or the perception and response to changes in bodily signals (such as heartbeat detection or satiety cues) [7].

Interoceptive experiences motivate behaviors that promote a return to homeostasis and are strongly linked to emotional and cognitive processing [8, 9]. Interoceptive signaling is shaped by neural predictions rooted in prior experiences and beliefs about bodily states. When these predictions are mismatched with the interoceptive signal being experienced, the resulting 'prediction error' putatively underlies various psychopathology, including eating disorders [6, 10]. Current cardiac interoceptive accuracy norms in healthy adolescents are comparable with those published in adults [11], while self-reported interoception continues to develop throughout adolescence [12, 13]. Domains such as trusting one's body appear to reduce as an individual moves through adolescence to adulthood [13]. Multidimensional batteries of interoception in a large sample of adolescents are lacking in the literature.

In AN, some studies have found that aberrant interoception emerges as a consequence of prolonged malnutrition, often lasting after weight restoration [14]. However, inconsistent reports of interoceptive functioning in AN across development require additional investigation [15–18]. The majority of these studies have been done in adults, with limited evidence of interoception in adolescents with AN. Studies of interoception in adolescent eating disorders transdiagnostically have found self-report differences in domains of interoception, where adolescents with EDs report greater body trust than adults with EDs [19]. Further, a network analysis demonstrated similar interoceptive capabilities between adults and adolescents with EDs, with lack of body trust being the most associated with eating disorder symptoms [20]. In a sample focused on children and adolescents, one study found no differences in self-reported interoception between AN and ARFID [21]. Additional research on interoception in adolescents with AN is needed to help shed light on these processes during a crucial period of development.

There are no systematic, multi-method data exploring interoceptive capabilities in adolescents with ARFID [7, 22, 23]. Clinically, individuals with ARFID report uncomfortable somatic experiences after eating, such as gastroesophageal reflux or gastric discomfort with satiety [24]. Thus, it is possible that disruptions in interoception may be a transdiagnostic feature of restrictive eating [21, 25]. In a review of interoceptive processes in eating disorders [6], Khalsa and colleagues report that while youth with ARFID may have elevated sensitivity in response to gastric signaling, whether this is driven by heightened sensitivity from external cues (e.g., smells and sounds of other eating) versus sensitivity to internal signaling (e.g., hunger cues) is unknown [26].

This proof-of-concept investigation sought to explore interoceptive differences between a healthy adolescent sample and two restrictive eating disorder samples: ARFID and AN. We also conducted a preliminary descriptive investigation on interoceptive differences between ARFID subgroups (sensory sensitivity, low appetite, fear of aversive consequences). Based on prior literature, we anticipated that adolescent norms would replicate existing adolescent data [27] and that youth with ARFID would demonstrate poorer interoceptive capabilities compared to HCs. Based on inconsistent literature on AN interoception, we anticipated that individuals with AN would have greater interoceptive accuracy than HC and ARFID but report less confidence and lower self-reported interoceptive capabilities. We had no a priori hypotheses about ARFID subgroup differences. We assessed interoception using a virtual heartbeat detection task (to probe for interoceptive accuracy, or how accurately one detects internal sensations such as heartbeat), confidence in heartbeat guesses (confidence scores correlated with accuracy scores provide a metric for *interoceptive awareness* or metacognitive awareness of interoceptive capabilities), and a self-report questionnaire on interoceptive abilities (*interoceptive sensibility*, or one's subjective judgment of their own perception of internal body signaling).

Methods

Participants

This exploratory study was conducted virtually during the COVID-19 pandemic and approved by Stanford University's Institutional Review Board. Participants who were 18 years of age provided consent; participants under 18 years of age provided assent, and their parents provided consent. Clinical participants were actively recruited from clinics, hospitals, schools, and through online advertisements. HC participants were recruited through online advertisements and schools. All participants were between the ages of 12-18. Clinical participants needed to have a diagnosis of ARFID or AN. Healthy comparison participants needed to be free of any psychiatric diagnoses. All participants completed a preliminary screen to determine eligibility. A larger sample of healthy control adolescents was collected (n=100)compared to the clinical samples (n=20-30) to replicate existing adolescent norms. Descriptive and clinical data for each group is provided in Table 1.

Study procedures

All study visits were done virtually since this study was completed during the COVID-19 pandemic. Before their visit, all participants were mailed a standard finger pulse oximeter to use during the heartbeat task (described below). A one-time 45-min Zoom appointment was scheduled to complete questionnaires, a diagnostic interview, and behavioral measures of interoception conducted by the first author.

Measures

We measured interoception via three domains, listed below, that are consistent with prior descriptions of interoception in the current literature:

Interoceptive accuracy [7, 27]: The ability to accurately detect internal body signals such as heart rate or hunger/ fullness cues.

Interoceptive awareness [27]: The association between one's confidence in or perception of their interoceptive accuracy and their actual interoceptive accuracy. In other words, a metacognitive awareness of interoceptive accuracy.

Table 1 Demographics and clinical features

	ARFID	AN	нс
Age M (SD) [†]	14.3 (1.9)	15.2 (1.9)	15.1 (1.8)
Sex (% female)	56.7%	95.2%	65.7%
Sex (% male)	43.3%	4.8%	34.3%
Gender			
% girls	46.7%	90.5%	56.6%
% boys	50%	4.7%	42.4
Prefer not to say	3.3%	4.8%	1%
Other	0%	0%	2%
Non-binary	3.3%	0%	0%
White	66.7%	57.1%	45.5%
Asian	3.3%	19%	22.2%
Black	-	-	4%
NHPI ^{††}	-	-	1%
Multi-racial	16.7%	19%	15.2%
Hispanic	16.7%	9.5%	12.1%
Comorbidity			
Major depressive disoder	23%	23%	-
Generalized anxiety disorder	20%	14%	-
Anxiety disorder NOS	3.33%	4.7%	-
Autism spectrum disorder	3.33%	-	-
Attention-deficit/hyperactivity disorder	20%	-	-
Psychotropic medication (% taking)	56.7%	23.8%	-
%EBW*	93.4 (14.2)	85.3 (15.3)	107.6 (17)
EDE-global	0.6 (0.9)	2.8 (1.6)	0.8 (1.0)
NIAS** sensory sensitivity profile	11.6 (3.5)	6.5 (4.1)	3.2 (3)
NIAS low appetite profile	8.9 (4.1)	4.7 (4.3)	2.7 (2.5)
NIAS fear profile	4.7 (4.0)	4 (3.8)	1.1 (1.9)

[†] M, Mean; SD, Standard deviation; ^{††}NHPI, Native Hawaiian and Pacific Islander; *EBW, Expected Body Weight; ** NIAS, Nine Item Avoidant/Restrictive Food Intake Disorder (ARFID) Screener—of note, clinical cut-offs for the ARFID screener subscales are as follows: (Sensory Sensitivity Profile \geq 10 & EDE-Q < 2.3; Low Appetite Profile \geq 9 & EDE-Q < 2.3; and Fear Profile \geq 10 & EDE-Q < 2.3) [31]

Interoceptive sensibility [7]: One's own perception (or self-report) of how tuned in they are to interoceptive signaling.

Cardiac interoception, as probed by heartbeat tasks, has been found to correlate with other measures of interoception interrogating other organ systems, such as sensitivity to gastrointestinal cues [7, 27], relevant for eating disorders. The tasks used to probe each domain are outlined below:

Demographics: Participants provided basic demographic information, including age, pubertal status, sex/gender, height, weight, ethnicity, comorbidities, and current medications. Percent of expected body weight (%EBW) was calculated using CDC median BMI for each participant's age and sex [28]. *Eating disorder assessment-5* [29]. The EDA-5 is a semistructured interview assessing the presence of a feeding/ eating disorder according to current DSM-5 criteria. This measure was used to confirm ED diagnosis for participants in the present study. A PhD-level clinician trained in the EDA-5 administered this assessment.

Nine-item ARFID screen [30]: The NIAS is a nineitem self-report questionnaire assessing eating habits in ARFID clinical presentations (fear, sensory sensitivity, and/or low appetite). Higher scores reflect greater clinical severity.

ARFID subgroups were determined by score cut-offs on the NIAS, established in the literature [31]. Specifically, to meet criteria for the sensory sensitivity subgroup, scores needed to $be \ge 10$ (sensitivity=0.97, specificity=0.63); to meet criteria for the low appetite subgroup, scores needed to $be \ge 9$ (sensitivity=0.86, specificity=0.70); and to meet criteria for the fear of aversive consequences subgroup, scores needed to $be \ge 10$ (sensitivity=0.68, specificity=0.89). The subgroups are defined as follows:

Sensory sensitivity: sensitivity to sensory characteristics of food, such as texture, temperature, or taste.

Low appetite: having a pervasive lack of interest in eating or foods.

Fear of aversive consequences: worry about an aversive consequence of eating, such as getting ill, choking, or vomiting.

Eating disorder examination-questionnaire [32]: The EDE-Q is a 32-item self-report questionnaire assessing eating disorder symptomatology over the past 28 days. Greater scores reflect greater severity of eating psychopathology driven by shape and weight concerns.

Heartbeat Tracking Task (Interoceptive Accuracy) [27, 33–35]. Participants silently counted their heartbeat (without manually checking) over six trials using randomly presented time windows of 25, 30, 35, 40, 45, and 50 s. Accuracy scores were calculated using: $1 - (|\text{nbeats}_{\text{real}} - \text{nbeats}_{\text{reported}}|)/(((\text{nbeats}_{\text{real}} + \text{nbeats}_{\text{reported}})/2)$, averaged over six trials, producing one accuracy score per participant. This score varies between 0 and 1, where scores of 1 indicate 100% accuracy in detecting one's heartbeat.

Confidence rating (interoceptive awareness) [27]. After each heartbeat trial, participants rated their confidence in the perceived accuracy of their response, using a continuous visual analog scale from 0 (total guess/no heartbeat awareness) to 10 (complete confidence/fully aware of heartbeat). An index of interoceptive awareness was computed using correlations between confidence ratings and accuracy scores, described above. Greater correlations reflect better interoceptive awareness.

The multidimensional assessment of interoceptive awareness-2 (MAIA-Y) (interoceptive sensibility) [13,

36]. The MAIA-Y is a 37-item questionnaire measuring domains of interoceptive awareness, where higher scores indicate greater attention to body signals. There are eight subscales: "noticing" (explores awareness of body sensations, i.e., "I can tell when I am uncomfortable in my body"); "not-distracting" (how tuned in on is to body sensations of discomfort, without using distraction); "notworrying" (index of participants' ability to not become emotionally reactive to negative physical sensations, i.e., "When I feel pain in my body I become upset"); "attention-regulation" (explores the capacity to regulate attention with multiple competing sensory stimuli, i.e., "I can focus on my entire body when I try"); "emotional-awareness" (explores the ability to be aware of the relationship between body-states and emotions, i.e., "I can feel how my body changes when I am angry"); "self-regulation" (explores the use of attention to body-states to regulate psychological distress, i.e., "I can use my breath to help me calm down and relax"); "body-listening" (index of attending to body sensations for insight, i.e., "I listen for clues from my body about my emotions"); and "trusting" (how much body cues are experienced as trustworthy, i.e., "I trust the way my body feels" or "I feel my body is a safe place"). Each subscale has a score from 0 to 5, where greater scores reflect more regulated and adaptive attention to body signaling or interoceptive cues.

Statistical approach

Data met assumptions of normality, and no outliers were identified in the dataset using visual examination of skewness and kurtosis and assessment of scatterplots to confirm the use of a GLM model. The pattern of missing data were assessed prior to analyses and identified to be missing completely at random (MCAR), by Little's MCAR test.

We conducted exploratory group comparisons to examine between-group effects. Considering the pilot nature of this study and small sample sizes of the clinical groups, we report partial eta square for the overall model, Hedge' g effect size estimates (ES) for pairwise comparisons, and Confidence Intervals [CI] for correlations [37–39]. For interpretation: η^2 of 0.01 = small effect, 0.06 = medium effect and 0.14 = large effect. For Hedge's g, 0.2 = small effect, 0.5 = medium effect, and 0.8 = large effect [40]. Analyses were completed using SPSS Version 27. A one-way between-subjects ANCOVA was conducted to compare interoceptive accuracy. A one-way ANCOVA was conducted to determine a statistically significant difference between groups (AN, HC, ARFID) on interoceptive sensibility (self-report subscales on the MAIA-Y) controlling for sex and %EBW. %EBW was added as a covariate in analyses since it varied between groups and can be implicated in heartbeat detection [41, 42]. Sex assigned at birth was also entered as a covariate in analyses as there have been reported differences in heartrate counting tasks between males and females across development [43]. Partial Pearson's correlations controlling for sex and %EBW were conducted to assess the degree of correlation between a participant's accuracy score and their level of confidence in their guess. Higher correlations reflect a greater awareness of internal body sensations, such as heartbeat. These data are not publicly available due to privacy and ethical restrictions.

We also descriptively explored interoceptive differences between ARFID subgroups. These differences are depicted via bar graphs (Figs. 1 and 2) and a table (Table 3).

Results

Interoceptive accuracy

A one-way ANCOVA was conducted to determine a statistically significant difference between groups (AN, HC, ARFID) on interoceptive accuracy (heartbeat detection) controlling for sex and %EBW. We found a significant effect of group on interoceptive accuracy F(2, 148)=4.71, η^2 =0.061 (medium effect size). For pairwise comparisons, individuals with ARFID had lower interoceptive accuracy than HC participants (M=0.66, SD=0.28], CI [0.02–0.29], Hedge's g=0.58, but not AN participants.

Interoceptive accuracy (Table 2) was greater for individuals with AN (M=0.69, SD=0.11) relative to those

with ARFID (M=0.50, SD=0.27; CI [-0.37 to -0.001], Hedge's g=0.86; F(2,149)=4.71, but not HC participants.

Interoceptive awareness

Interoceptive awareness was greater for HC (r(98) = 0.35 (medium effect), CI [0.16–0.52]) and ARFID participants (r(28) = 0.40 (medium effect), CI [0.10–0.68]) relative to AN participants (r(20) = 0.13 (small effect), CI [-0.31 to 0.54]).

Interoceptive sensibility

We found a significant effect of group on specific MAIA-Y subscales, detailed below.

ARFID. For pairwise comparisons, participants with ARFID reported lower scores on the "Not Worry" subscale, reflecting more difficulty not becoming emotionally reactive to negative physical sensations (F(2, 149)=4.71) compared to HC (Hedge's g=0.65, CI [-0.11 to 0.1.00]) but not AN. Participants with ARFID also had lower scores on the "Self-Regulation" and "Body Trust" subscale compared to HC, but not AN, indicating greater difficulty in using attention to body states to self-regulate physiological distress and less trust in body sensations (F(2,149)=5.91; Hedge's g=0.70-0.78, CI [-0.12 to 1.18]).

AN. Participants with AN reported lower scores than HC participants but not ARFID on the "Noticing" subscale (Hedge's g=0.74), the "Not Distract" subscale (Hedge's g=0.58), and the "Self-Regulation" subscale



Fig. 1 Interoceptive accuracy by ARFID subgroup



Fig. 2 Interoceptive sensibility by ARFID subgroup

Table 2 Main Interoceptive Outcomes

Interoceptive domain M (SD)	ARFID n=30	AN n=22	HC n=100	SS	F	95% CI	Hedge's g
Interoceptive accuracy	0.50 (0.27)	0.69 (0.11)	0.66 (0.28)	0.62	4.71	-0.37 to 0.001	ARFID < AN = 0.86
						-0.22 to 0.29	ARFID < HC = 0.47
Interoceptive awareness (Pear- son's r[95% CI])	0.40 [0.10 to 0.68]	0.13 [-0.31 to 0.54]	0.35 [0.16–0.51]	-	-	-	-
MAIA-Y noticing	3.14 (0.96)	2.73 (0.86)	3.36 (0.84)	5.49	3.70	0.07-1.2	AN < HC = 0.74
MAIA-Y not distract	2.10 (1.03)	1.87 (0.93)	2.54 (0.96)	7.97	4.24	0.03-1.30	AN < HC = 0.80
MAIA-Y not worry	1.94 (0.84)	2.19 (0.83)	2.50 (0.85)	6.59	4.71	0.11-1.00	ARFID < HC = 0.65
MAIA-Y attention regulation	2.40 (1.02)	2.30 (0.81)	2.76 (0.91)	4.38	2.59		All < 0.2
MAIA-Y emotion regulation	291 (1.10)	2.65 (1.03)	2.99 (0.93)	1.60	0.83		All < 0.2
MAIA-Y self-regulation	1.91 (1.02)	1.90 (1.34)	2.56 (0.95)	11.59	5.91	0.12-1.18	ARFID < HC = 0.70
						0.02-1.34	AN < HC = 0.58
MAIA-Y body listening	1.76 (1.20)	1.25 (0.99)	2.17 (1.01)	12.40	5.72	0.24-1.62	AN < HC = 0.92
MAIA-Y body trust	2.99 (1.10)	1.82 (1.38)	3.77 (0.98)	53.64	24.6	- 1.26 to 2.6	AN < HC = 1.84
						- 1.91 to 0.41	AN < ARFID = 0.95
						- 1.35 to 0.23	$ARFID\!<\!HC\!=\!0.78$

Hedge's g, 0.2 = small effect; 0.5 = medium effect; 0.8 = large effect; Pearson's r, 0.1 = small effect; 0.3 = medium effect; > 0.5 = large effect

 Table 3
 Interoceptive awareness between ARFID subgroups

Subgroup	r	95% [CI]
Sensory sensitivity (n = 12)	0.55	-0.04 to 0.85
Low appetite (n=4)	0.27	-0.93 to 0.98
Sensory sensitivity + low appetite (n = 8)	0.49	-0.33 to 0.89
Low appetite + fear $(n = 1)$	-	-
Sensory sensitivity + low appetite + fear (n = 5)	0.26	-0.81 to 0.93

Pearson's r reflecting correlation between accuracy scores and confidence of accuracy guess

(Hedge's g=0.58), reflecting that individuals with AN are more likely to notice and distract themselves from uncomfortable/painful body sensations, and are less likely to use self-regulation strategies. Participants with AN also reported lower scores on the "Body Listening" subscale compared to HC participants only, reflecting they are less likely to listen to their body signals/cues, (F(2,149)=5.72; Hedge's g=0.92, CI [0.24–1.62]). Lastly, participants with AN reported lower scores on the "Trusting" subscale indicating difficulty experiencing their bodies as trustworthy (F(2,149)=24.64; Hedge's g > 0.80 compared to both ARFID (CI [–1.91 to 0.41]) and HC participants (CI [–1.26 to 2.6]).

ARFID subgroup

In the entire ARFID sample (n = 30), no participants fell in the "Fear" subgroup exclusively. We used the NIAS and validated scoring guidelines and clinical cut-offs to assign group membership [31]. Our group membership was as follows: 12 participants in the "Sensory Sensitivity Eating" subgroup, 4 participants in the "Low Appetite" subgroup, 8 participants in the "Sensory Sensitivity Eating + Low Appetite" subgroup, 1 participant in the "Low Appetite + Fear of Aversive Consequences" subgroup, and 5 participants in all three subgroups. The Low Appetite subgroup had an interoceptive profile with the greatest accuracy (M = 0.83, SD = 0.14) and lowest awareness (r = 0.27, [-0.93 to 0.98]). The Sensory Sensitivity Eating subgroup appeared more consistent with the overall ARFID group, but this may also be because of the relatively larger sample (Accuracy M=0.45 SD=0.22; Awareness r=0.55 [-0.0 to 85]). Descriptive findings for all groups can be found in Fig. 1 and Table 3. For interoceptive sensibility, Body Trust was the highest in the Sensory Sensitivity and Low Appetite Groups, while Self-Regulation was lowest in the Low Appetite + Fear of Aversive Consequences Group (Fig. 2). Tests of significance or effect sizes were not run or generated due to small sample size and unequal distribution.

Discussion

Overall, the results of this pilot study begin to shed light on interoceptive processes in adolescents with a restrictive eating disorder (AN and ARFID) and healthy comparisons. These data replicate existing adolescent norms using a multi-method battery of interoception and explore differences between two restrictive eating disorders for whom interoceptive functioning may relate to symptom development and expression [6]. In our sample, heartbeat guessing accuracy in healthy adolescents was found to be comparable to a sample of 80 adults completing the same task [27], consistent with prior research suggesting interoceptive accuracy may be established by adolescence [11]. Further, data show interoceptive awareness was slightly greater in a normative adult sample (r = 0.56) than in our adolescent sample (r = 0.35), indicating awareness of body signaling may continue to develop throughout adolescence. Lastly, for interoceptive sensibility in HC adolescents, our sample replicated the norms reported in the original validation study, highlighting that interoceptive sensibility continues to develop through adolescence [13].

With regards to interoceptive accuracy in clinical samples, it appeared that adolescents with ARFID *did* have poorer interoceptive accuracy (heartbeat detection) than HC participants, consistent with a priori hypotheses. However, ARFID participants' confidence levels aligned with their heartbeat guesses, similar to HCs, suggesting that ARFID participants have intact metacognitive awareness of their interoceptive accuracy compared to AN participants, and similar to their healthy adolescent counterparts. These data may shed light on ARFID's clinical picture: awareness of one's reduced interoceptive accuracy may lead to increased caution around various aspects of eating, such as swallowing, eating quickly, or trying new foods.

Adolescents with AN had relatively higher interoceptive accuracy scores than those with ARFID (comparable to HC), which is partially consistent with our a priori hypotheses. However, their confidence did *not* map onto their accuracy, suggesting awareness of internal body sensations was relatively lower in AN participants relative to ARFID and HC groups. Prior research on interoceptive accuracy in AN has focused on adult samples, producing equivocal results [15, 44]. Consistent with our findings, one study of adults with AN reported no differences in heartbeat-tracking accuracy between AN participants and HCs, but found that participants with AN had lower interoceptive awareness [18].

Interoceptive sensibility, or self-reported interoception, on some subscales of the MAIA-Y varied between clinical cohorts and HC participants, aligned with hypotheses. Specifically, participants with ARFID reported

difficulty not worrying about negative body sensations, using attention to body states to self-regulate distress, and trusting body signals. These findings are consistent with prior reports of ARFID participants viewing internal body sensations such as hunger or fullness as frightening, untrustworthy, or aversive [45] and identify a gap in this population's ability to make sense of body cues and act adaptively. Individuals with AN reported a greater likelihood of ignoring uncomfortable body sensations and a lower likelihood of listening to or trusting their body cues. These findings may partially explain why interoceptive awareness was lower during the heartbeat guessing task: despite having relatively good accuracy, participants with AN have difficulty trusting and listening to body cues such as heartbeat-or more relevant to AN itselfgastric cues of hunger/satiety. However, individuals with AN may selectively attend to illness-specific cues, such as fullness and satiety, complicated by post-prandial discomfort in the context of slower digestion and gastric emptying [46].

Lastly, exploratory descriptive analyses within the ARFID group found relatively greater interoceptive accuracy in the Low Appetite subgroup than in other subgroups. The Low Appetite subgroup also had the lowest interoceptive awareness. This interoceptive profile looks similar to the AN group (e.g., greater accuracy and lower awareness). The exclusively Sensory Sensitivity subgroup had the lowest accuracy but the highest awareness, more characteristic of the ARFID subgroup as a whole. Regarding interoceptive sensibility, the Low Appetite and Sensory Sensitivity subgroups had greater body trust. Greater body trust may distinguish participants with Low Appetite ARFID participants from participants with AN, despite other domains of interoception looking similar. As prior research has shown, body trust reduces with age and negatively correlates with eating disorder symptoms. Thus, fostering body trust in adolescence may be a protective factor transdiagnostically [13, 19, 20]. Given the small sample size and unequal distributions, these interpretations should be read with caution.

This pilot study should be considered in light of several important limitations. First, due to this project's scope and limited funding, the first author completed all the assessments and was not masked to the diagnostic group. Future investigations benefit from masking assessors to reduce potential bias. Second, given ARFID's heterogenous diagnostic categorization [3], interoception capabilities may vary by clinical presentation (fear, low appetite, and/or sensory sensitivity). We provide preliminary data probing ARFID subgroups; however, future research should consider larger samples to power investigations of interoception across different clinical presentations of ARFID. Next, as mentioned, the heartbeat tracking task can be influenced by factors such as BMI [47, 48] and sex assigned at birth [43], though both %EBW and sex were controlled for in current analyses. Future designs may be strengthened by matching participants for gender and sex. The heartbeat tracking task was used in this proofof-concept study because it is a practical, low-cost, low-burden assessment procedure to interrogate interoception. Lastly, we recognize that the clinical sample sizes are small and preclude robust hypothesis testing. However, the results suggest that adolescents with restrictive eating disorders have differential interoceptive capabilities than healthy comparisons.

Conclusions

The preliminary data reported here add to our understanding of interoception broadly and within restrictive eating disorders such as ARFID and AN. While this study is the first to our knowledge to explore interoceptive capabilities explicitly in youth with ARFID using a multi-method battery and exploring by subgroup, prior data suggest that these processes may be disrupted in this population. For example, painful and uncomfortable somatic sensations are common in children with ARFID, along with sensitivity to digestive cues [24, 49]. Future studies may consider using an interoceptive assessment that probes gastric cues, such as the water-loading task [50], for a more comprehensive and reliable interrogation of this population's interoceptive processes related to eating. The pilot data presented here are a first step in understanding how body sensations are understood and perceived in adolescent restrictive eating disorders compared to HCs.

Abbreviations

ARFID Avoidant/restrictive food intake disorder

- AN Anorexia nervosa
- HC Healthy controls
- ED Eating disorder
- ES Effect size
- CI Confidence interval

MAIA-Y The multidimensional assessment of interoceptive awareness-2

Acknowledgements

The authors thank all the participants for their contribution to this pilot project!

Author contributions

Both authors contributed to the study conception and design. Material preparation, data collection, and analyses were performed by ND. The first draft of the manuscript was written by ND, and JL commented on previous versions of the manuscript. All authors read and approved the final manuscript.

Funding

Stanford's Maternal Child Health Research Institute Post-Doctoral Award FY21 (salary support for author N.D.).

Availability of data and materials

The datasets generated and analyzed during the current study are not publicly available due to privacy and ethical reasons but are available from the corresponding author on reasonable request.

Declarations

Ethics approval and consent to participate

All participants signed consent/assent forms for participation in this project. The project received certification of Human Subject's Approval by Stanford's Institutional Review Board, eProtocol #54246.

Consent for publication

Not applicable.

Competing interests

The authors have no relevant financial or non-financial interests or competing interests to disclose.

Author details

¹Department of Psychiatry and Behavioral Sciences, Stanford University School of Medicine, 401 Quarry Road, Stanford, CA 94305, USA.

Received: 28 April 2023 Accepted: 17 October 2023 Published online: 23 October 2023

References

- Association AP. Diagnostic and statistical manual of mental disorders. 5th ed. 2013.
- Becker KR, Keshishian AC, Liebman RE, Coniglio KA, Wang SB, Franko DL, et al. Impact of expanded diagnostic criteria for avoidant/restrictive food intake disorder on clinical comparisons with anorexia nervosa. Int J Eat Disord. 2019;52(3):230–8.
- Thomas JJ, Lawson EA, Micali N, Misra M, Deckersbach T, Eddy KT. Avoidant/restrictive food intake disorder: a three-dimensional model of neurobiology with implications for etiology and treatment. Curr Psychiatry Rep. 2017;19(8):1–9.
- Fisher MM, Rosen DS, Ornstein RM, Mammel KA, Katzman DK, Rome ES, et al. Characteristics of avoidant/restrictive food intake disorder in children and adolescents: a new disorder in DSM-5. JAH J Adolescent Health. 2014;55(1):49–52.
- Sanchez-Cerezo J, Nagularaj L, Gledhill J, Nicholls D. What do we know about the epidemiology of avoidant/restrictive food intake disorder in children and adolescents? A systematic review of the literature. Eur Eat Disord Rev. 2022;31:226–46.
- Khalsa SS, Berner LA, Anderson LM. Gastrointestinal interoception in eating disorders: charting a new path. Curr Psychiatry Rep. 2022;24(1):47–60.
- Khalsa SS, Adolphs R, Cameron OG, Critchley HD, Davenport PW, Feinstein JS, et al. Interoception and mental health: a roadmap. Biol Psychiatry Cognit Neurosci Neuroimaging. 2018;3(6):501–13.
- Wiens S. Interoception in emotional experience. Curr Opin Neurol Curr Opin Neurol. 2005;18(4):442–7.
- Tsakiris M, Critchley H. Interoception beyond homeostasis: affect, cognition and mental health. Philos Trans Roy Soc B Biol Sci. 2016;371(1708):20160002.
- 10. Paulus MP. Interoception in anxiety and depression. 2010.
- Mai S, Wong CK, Georgiou E, Pollatos O. Interoception is associated with heartbeat-evoked brain potentials (HEPs) in adolescents. Biol Psychol. 2018;137:24–33.
- 12. Todd J, Aspell JE, Barron D, Swami V. An exploration of the associations between facets of interoceptive awareness and body image in adolescents. Body Image. 2019;31:171–80.
- Jones A, Silas J, Todd J, Stewart A, Acree M, Coulson M, et al. Exploring the multidimensional assessment of interoceptive awareness in youth aged 7–17 years. J Clin Psychol. 2021;77(3):661–82.

- Khalsa SS, Craske MG, Li W, Vangala S, Strober M, Feusner JD. Altered interoceptive awareness in anorexia nervosa: effects of meal anticipation, consumption and bodily arousal. Int J Eat Disord. 2015;48(7):889–97.
- Pollatos O, Kurz A-L, Albrecht J, Schreder T, Kleemann AM, Schopf V, et al. Reduced perception of bodily signals in anorexia nervosa. Eat Behav. 2008;9(4):381–8.
- Lutz APC, Schulz A, Voderholzer U, Koch S, van Dyck Z, Vogele C. Enhanced cortical processing of cardio-afferent signals in anorexia nervosa. Clin Neurophysiol. 2019;130(9):1620–7.
- Richard A, Meule A, Georgii C, Voderholzer U, Cuntz U, Wilhelm FH, et al. Associations between interoceptive sensitivity, intuitive eating, and body mass index in patients with anorexia nervosa and normal-weight controls. Eur Eat Disord Rev. 2019;27(5):571–7.
- Kinnaird E, Stewart C, Tchanturia K. Interoception in anorexia nervosa: exploring associations with alexithymia and autistic traits. Front Psychiatry. 2020;11:64.
- Brown TA, Berner LA, Jones MD, Reilly EE, Cusack A, Anderson LK, et al. Psychometric evaluation and norms for the multidimensional assessment of interoceptive awareness (MAIA) in a clinical eating disorders sample. Eur Eat Disord Rev. 2017;25(5):411–6.
- Brown TA, Vanzhula IA, Reilly EE, Levinson CA, Berner LA, Krueger A, et al. Body mistrust bridges interoceptive awareness and eating disorder symptoms. J Abnorm Psychol. 2020;129(5):445–56.
- Zanna V, Criscuolo M, Mereu A, Cinelli G, Marchetto C, Pasqualetti P, et al. Restrictive eating disorders in children and adolescents: a comparison between clinical and psychopathological profiles. Eat Weight Disord Stud Anorexia Bulimia Obes. 2021;26(5):1491–501.
- Khoury NM, Lutz J, Schuman-Olivier Z. Interoception in psychiatric disorders: a review of randomized, controlled trials with interoception-based interventions. Harv Rev Psychiatry. 2018;26(5):250–63.
- 23. Herbert PO. The interoceptive basis of the mind. In: Manos Takiris HDP, editor. Oxford University Press; 2018.
- Sharp WG, Stubbs KH, Adams H, Wells BM, Lesack RS, Criado KK, et al. Intensive, manual-based intervention for pediatric feeding disorders: results from a randomized pilot trial. J Pediatr Gastroenterol Nutr. 2016;62(4):658–63.
- Zucker NL, LaVia MC, Craske MG, Foukal M, Harris AA, Datta N, et al. Feeling and body investigators (FBI): ARFID division—an acceptance-based interoceptive exposure treatment for children with ARFID. Int J Eat Disord. 2019;52(4):466–72.
- Frank GKW, Golden NH, Murray HB. Introduction to a special issue on eating disorders and gastrointestinal symptoms—the chicken or the egg? Int J Eat Disord. 2021;54(6):911–2.
- Garfinkel SN, Seth AK, Barrett AB, Suzuki K, Critchley HD. Knowing your own heart: distinguishing interoceptive accuracy from interoceptive awareness. Biol Psychol. 2015;104:65–74.
- Le Grange DD, Swanson SA, Ludwig K, Glunz C, Kreipe RE. Calculation of expected body weight in adolescents with eating disorders. Pediatrics. 2012;129(2):e438–46.
- Sysko R, Glasofer DR, Hildebrandt T, Klimek P, Mitchell JE, Berg KC, et al. The eating disorder assessment for DSM-5 (EDA-5): development and validation of a structured interview for feeding and eating disorders. Int J Eat Disord. 2015;48(5):452–63.
- Zickgraf HF, Ellis JM. Initial validation of the nine item avoidant/restrictive food intake disorder screen (NIAS): a measure of three restrictive eating patterns. Appetite. 2018;123:32–42.
- Burton Murray H, Dreier MJ, Zickgraf HF, Becker KR, Breithaupt L, Eddy KT, et al. Validation of the nine item ARFID screen (NIAS) subscales for distinguishing ARFID presentations and screening for ARFID. Int J Eat Disord. 2021;54(10):1782–92.
- Fairburn CG, Beglin SJ. Assessment of eating disorders—interview or selfreport questionnaire. Int J Eat Disord. 1994;16(4):363–70.
- Hart N, McGowan J, Minati L, Critchley HD. Emotional regulation and bodily sensation: interoceptive awareness is intact in borderline personality disorder. J Pers Disord. 2013;27(4):506–18.
- Herbert BM, Muth ER, Pollatos O, Herbert C. Interoception across modalities: on the relationship between cardiac awareness and the sensitivity for gastric functions. PLoS ONE. 2012;7(5):e36646.
- 35. Schandry R. Heart beat perception and emotional experience. Psychophysiology. 1981;18(4):483–8.

- Mehling WE, Gopisetty V, Daubenmier J, Price CJ, Hecht FM, Stewart A. Body awareness: construct and self-report measures. PLoS ONE. 2009;4(5):e5614.
- 37. HC K. Effect size. RLCSO L, editor2014. p. 1-3.
- 38. Kraemer HC. Is it time to ban the P value? JAMA Psychiat. 2019;76(12):1219–20.
- Kraemer HC, Neri E, Spiegel D. Wrangling with p-values versus effect sizes to improve medical decision-making: a tutorial. Int J Eat Disord. 2020;53(2):302–8.
- Cohen J. Statistical power analysis for the behavioral sciences. Routledge; 1977.
- Ring C, Brener J. Heartbeat counting is unrelated to heartbeat detection: a comparison of methods to quantify interoception. Psychophysiology. 2018;55(9):e13084.
- Desmedt O, Luminet O, Corneille O. The heartbeat counting task largely involves non-interoceptive processes: evidence from both the original and an adapted counting task. Biol Psychol. 2018;138:185–8.
- Prentice F, Murphy J. Sex differences in interoceptive accuracy: a metaanalysis. Neurosci Biobehav Rev. 2022;132:497–518.
- Pollatos O, Herbert BM, Berberich G, Zaudig M, Krauseneck T, Tsakiris M. Atypical self-focus effect on interoceptive accuracy in anorexia nervosa. Front Hum Neurosci. 2016;10:484.
- 45. Zickgraf HF, Loftus P, Gibbons B, Cohen LC, Hunt MG. "If I could survive without eating, it would be a huge relief": development and initial validation of the Fear of Food Questionnaire. Appetite. 2022;169:105808.
- 46. Heruc GA, Little TJ, Kohn MR, Madden S, Clarke SD, Horowitz M, et al. Effects of starvation and short-term refeeding on gastric emptying and postprandial blood glucose regulation in adolescent girls with anorexia nervosa. Am J Physiol Endocrinol Metab. 2018;315(4):E565–73.
- Rouse CH, Jones GE, Jones KR. The effect of body-composition and gender on cardiac awareness. Psychophysiology. 1988;25(4):400–7.
- Knapp-Kline K, Kline JP. Heart rate, heart rate variability, and heartbeat detection with the method of constant stimuli: slow and steady wins the race. Biol Psychol. 2005;69(3):387–96.
- Cooney M, Lieberman M, Guimond T, Katzman DK. Clinical and psychological features of children and adolescents diagnosed with avoidant/ restrictive food intake disorder in a pediatric tertiary care eating disorder program: a descriptive study. J Eat Disord. 2018;6:1–8.
- Brown TA, Perry TR, Kaye WH, Wierenga CE. Pilot study of a water load test as a measure of gastric interoception in anorexia nervosa. Eat Weight Disord. 2022;27(6):2223–8.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

